Response to Comments of Reviewer & Scientific Editor

Reviewer 1
Comment: Interesting manuscript that aimed to review the anti-inflammatory and immunomodulatory potential of omega-3 long-chain polyunsaturated fatty acids (ω-3 LCPUFA) and its plausible therapeutic as a pharmaconutrition intervention strategy in a management of inflammatory cytokine storm associated with COVID-19, with special reference to NAFLD. Manuscript is with a new subject and well writing. Missing some metabolic pathways of cytokines storm, one of the important manuscript subject.

Response: Metabolic pathway involved in the cytokine storm is included in the revised version.

Scientific Editor:
Comment 1: The title is too long, and it should be no more than 18 words
Response: Revised the title accordingly.

Comment 2: The “Author Contributions” section is missing. Please provide the author contributions
Response: Included in the revised version

Comment 3: The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor; and
Response: Yes, now we have included the original picture of the manuscript.

Comment 4: If an author of a submission is re-using a figure or figures published elsewhere, or that is copyrighted, the author must provide documentation that the previous publisher or copyright holder has given permission for the figure to be re-published; and correctly indicating the reference source and copyrights.
Response: No, it is drawn for this article.